Author's response to reviews

Title: Promoting prudent use of antibiotics: the experience from a multifaceted regional campaign in Greece

Authors:

Diamantis Plachouras (dplach@gmail.com)
Anastasia Antoniadou (ananto@med.uoa.gr)
Efthimia Giannitsioti (gianiem@hotmail.com)
Lambrini Galani (e.giamarellou@hygeia.gr)
Ioannis Katsarolis (ikatsarolis@yahoo.com)
Dimitra Kavatha (dimitra.kavatha@gmail.com)
George Koukos (georgekoukos78@yahoo.gr)
Periklis Panagopoulos (ppanag@med.uoa.gr)
Antonios Papadopoulos (antpapa1@otenet.gr)
Garyfallia Poulakou (gpoulakou@gmail.com)
Vissaria Sakka (vissaria@gmail.com)
Maria Souli (msouli@med.uoa.gr)
Styliani Symbardi (ikatsarolis@gmail.com)
Sotirios Tsiodras (sotirios.tsiodras@gmail.com)
Kyriaki Kanellakopoulou (kyrkanel@yahoo.com)
Helen Giamarellou (e.giamarellou@hygeia.gr)

Version: 4 Date: 19 December 2013

Author's response to reviews: see over
RESPONSE TO REVIEWER 1

Point 2 - replacement done

Point 8 - reference nº 3) placed immediately following the statement.

Point 17 - this information was added in the manuscript

"It must be stressed that antibiotics are prescription-only drugs in Greece. However this is not strictly enforced, potentially leading to over the counter antimicrobial sales by pharmacies."

Point 18 - adverse effects reported by the respondents were added as a reference under the relevant table 1

"Adverse events reported: 31 (35.6%) rash, 26 (29.9%) allergic reaction, 16 (18.4%) diarrhea and 14 (16%) combination of these."

Point 19 - A table was added (Table 3) in order to address the comments made by the reviewer. P was italicized where appropriate and all values were reported numerically.